Author's response to reviews

Title: Using Ontario's "Telehealth" Health Telephone Helpline as an Early-Warning System: A Study Protocol

Authors:

Elizabeth Rolland (erolland@gmail.com)
Kieran M Moore (moorek1@kgh.kari.net)
Victoria A Robinson (victoria.robinson@utoronto.ca)
Don McGuinness (dmcguinness@healthunit.on.ca)

Version: 3 Date: 27 January 2006

Author's response to reviews:

As per your request, the following revisions were made:
- Title: title has been modified as per your requirements
- Abstract now includes subtitles and has also been included in the actual manuscript.
- Tables: as the table is less than 2 pages long, it has been incorporated into the manuscript - it can be found at the end of the manuscript.